\documentclass[11pt]{article}
\usepackage{fullpage} % 1 inch margins on standard letter size paper
\usepackage{amsmath}  % AMS math macros
\usepackage{amssymb}  % Additional math symbols from AMS

\title{Assignment 0}
\author{Your name here}

\begin{document}
\maketitle

Enter your two solutions below, typeset as ``display style'' formulas,
like this:
\[ a x^2 + b x + c \]

\begin{enumerate}
\item \textsl{Enter the formula corresponding to the initial letter of
  your first name below.}

  Replace with your answer
  
\item \textsl{Enter the formula corresponding to the initial letter of
  your last name, or the ``Alt'' formula if this is the same letter as
  your first name, below.}

  Replace with your answer
  
\end{enumerate}

\end{document}
